Author's response to reviews

Title: Mobile phones are a viable option for surveying young Australian women: a comparison of two telephone survey methods

Authors:

Bette Liu (bliu@kirby.unsw.edu.au)
Julia ML Brotherton (jbrother@vcs.org.au)
David Shellard (david@hvrf.com.au)
Basil Donovan (bdonovan@kirby.unsw.edu.au)
Marion Saville (msaville@vcs.org.au)
John M Kaldor (jkaldor@unsw.edu.au)

Version: 4 Date: 22 November 2011

Author's response to reviews:

Dear Editorial team,

Please find that additional formatting changes made in the attached manuscript.

Kind Regards,
Bette